Author's response to reviews

Title: Health impact assessment of waste management facilities in three European countries

Authors:

Francesco Forastiere (forastiere@asplazio.it)
Chiara Badaloni (badaloni@asplazio.it)
Kees de Hoogh (c.dehoogh@imperial.ac.uk)
Martin K von Kraus (MAK@euro.who.int)
Marco Martuzzi (mam@ecr.euro.who.int)
Francesco Mitis (mit@ecr.euro.who.int)
Lubica Palkovicova (lubica.palkovicova@szu.sk)
Daniela Porta (porta@asplazio.it)
Philipp Preiss (philipp.preiss@ier.uni-stuttgart.de)
Andrea Ranzi (aranzi@arpa.emr.it)
Carlo A Perucci (perucci@agenas.it)
David Briggs (d briggs@imperial.ac.uk)

Version: 5 Date: 12 May 2011

Author's response to reviews: see over
Reviewer comment
The authors have responded to previous comments sufficiently. Noted some minor spelling/language errors in some of the updated text (e.g. new discussion of confidence levels in HIA).
Done (all the changes are in revision mode)

In addition, we suggest that you rewrite the Conclusions section (also in the abstract), as it appears too descriptive, without summarizing the inference that can be extracted from your work.
We have slightly modified both.

We normally request that manuscript titles include the study design, for example "A versus B in the treatment of C: a randomized controlled trial X is a risk factor for Y: a case control study", perhaps by moving the first three words to the end. If this is not appropriate for some reason, then we can stick with the title as is.
We believe that the title is appropriate as it is.

On the title page, the $ should be removed in the heading *Corresponding author.
Done

The periods after the headings in the abstract should be removed and the [0] subscript at the end of the Abstract's Conclusion section should be removed.
Done

Periods are needed after the abbreviation et al. throughout the manuscript.
Done

At the bottom of page 17 a space is needed between 2001 and was.
Done

The underlining on pages 26 and 27 should be removed and changed to normal text.
Done

The heading after Competing interests should read Authors' contributions; and please check reference #4. Also in the references section, Copenhagen is spelled incorrectly in references 1 and 2.
Done

In the tables all horizontal lines must be visible.
Done